Author's response to reviews

Title: Toxoplasmosis presenting as a swelling in the axillary tail of the breast and a palpable axillary lymph node mimicking malignancy: A case report.

Authors:

HP Priyantha Siriwardana (hppsir@hotmail.com)
Louise Teare (louise.teare@meht.nhs.uk)
Dia Kamel (dia.kamel@meht.nhs.uk)
Reggie Inwang (reggie.inwang@meht.nhs.uk)

Version: 5 Date: 16 May 2011

Author's response to reviews: see over
Dear Casy,

MS: 5109294574767753. TOXOPLASMOSIS PRESENTING AS A SWELLING IN THE AXILLARY TAIL OF THE BREAST AND A PALPABLE AXILLARY LYMPH NODE MIMICKING MALIGNANCY: A CASE REPORT

Thank you for your e-mail today saying that the Current Science Editorial Office has completed the manuscript checks on our submission and advising to address the following concerns in a revised manuscript:

1. There is no consent section in the manuscript and this should be added according to the instructions for authors.

Consent section is now added to the revised manuscript. (Page 9)

2. I don't believe the age of the children in this case is relevant and perhaps this should be excluded from the case.

Yes, we agree and it is now excluded in the revised manuscript.

Thanking You

Yours sincerely

HP Priyantha Siriwardana MS FRCS
Corresponding author.